Author's response to reviews

Title: An Electronic Application for Rapidly Calculating Charlson Comorbidity Score

Authors:

William H Hall III (william.hall@ucdmc.ucdavis.edu)
Ramanathan Ramachndran (ramanath_r1@yahoo.com)
Samir Narayan (samir.narayan@ucdmc.ucdavis.edu)
Ashesh B Jani (jani_1969@yahoo.com)
Srinivasan Vijayakumar (vijay@ucdavis.edu)

Version: 3 Date: 22 October 2004

Author's response to reviews: see over
October 21, 2004

Iraxte Puebla
Assistant Editor
BioMed Central
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Dear Ms. Puebla:

I am writing in response to your e-mail of yesterday, reminding me about making changes to our manuscript, "An Electronic Application for Rapidly Calculating Charlson Comorbidity Score" (MS# 2096665906334100). As requested, we have made the following formatting changes:

● In the manuscript text, placed the reference before the full stop.
● Reviewed the new policy on Competing Interests, and declared that we have no competing interests. This declaration is on the title page.

The attached manuscript has those changes.

Thank you for being in touch with us, and we look forward to seeing our manuscript published.

Sincerely,

Srinivasan Vijayakumar

Srinivasan Vijayakumar, M.D., F.A.C.R.
Professor and Chair
Department of Radiation Oncology
University of California, Davis